Author's response to reviews

Title: Factors affecting antiretroviral pharmacokinetics in HIV-infected women with virologic suppression on combination antiretroviral therapy: a cross-sectional study

Authors:

Mona R. Loutfy (mona.loutfy@wchospital.ca)
Sharon L. Walmsley (sharon.walmsley@uhn.on.ca)
Marina B. Klein (marina.klein@mcgill.ca)
Janet Raboud (janet@raboud.net)
Alice L. Tseng (alice.tseng@uhn.on.ca)
Sandra L. Blitz (sandra.blitz@uhnresearch.ca)
Neora Pick (npick@cw.bc.ca)
Brian Conway (bconway@exchange.ubc.ca)
Jonathan B. Angel (jangel@ohri.ca)
Anita R. Rachlis (Anita.Rachlis@sunnybrook.ca)
Kevin Gough (goughk@smh.toronto.on.ca)
Jeff Cohen (jcohenmd@cogeco.ca)
David Haase (david.haase@cdha.nshealth.ca)
David Burdge (david.burdge@gmail.com)
Fiona M. Smaill (smaill@mcmaster.ca)
Alexandra de Pokomandy (alexandra.depokomandy@mcgill.ca)
Hugues Loemba (hloemba@uohs.uottawa.ca)
Sylvie Trottier (sylvie.trottier@crchul.ulaval.ca)
Charles J. la Porte (cjl.laporte@gmail.com)

Version: 2 Date: 6 May 2013

Author's response to reviews:

Comment #1: Ethics Statement
Response: An Ethics Statement paragraph has now been included under Methods Section.

Comment #2: Name of Ethics Committee
Response: The name of the Ethics Committee was added.

Comment #3: Full Role of SW, MBK, NP, BC, JBA, ARR, KG, JC, DH, DB, FMS, AdP, HL, ST
Response: The full role of these co-authors has now been included in the revised manuscript